Author's response to reviews

Title: Building capacity for evidence generation, synthesis and implementation to improve the care of mothers and babies in South East Asia: methods and design of the SEA-ORCHID Project using a logical framework approach

Authors:

Steve McDonald (steve.mcdonald@med.monash.edu.au)
Tari Turner (tari.turner@med.monash.edu.au)
Catherine Chamberlain (chamberl@ihug.com.au)
Pisake Lumbiganon (pisake@kku.ac.th)
Jadsada Thinkhamrop (jadsada@kku.ac.th)
Mario R Festin (mfestinmd@yahoo.com)
Jacqueline J Ho (jho@pc.jaring.my)
Hakimi Mohammad (hakimi@chnrl.net)
David J Henderson-Smart (dhs@mail.usyd.edu.au)
Jacki Short (jacshort@hotmail.com)
Caroline A Crowther (caroline.crowther@adelaide.edu.au)
Ruth Martis (ruth-martis@clear.net.nz)
Sally Green (sally.green@med.monash.edu.au)
the SEA-ORCHID Study Group (steve.mcdonald@med.monash.edu.au)

Version: 4 Date: 30 June 2010

Author's response to reviews:

As requested, I have added a comma on Page 10 (second paragraph) between Philippines and Manila. No other changes were requested.